Author's response to reviews

Title: Oral and Anal Sex Practices among high School Youth in Addis Ababa, Ethiopia.

Authors:

Amsale Cherie cherie (amsalec2002@yahoo.com)
Yemane Berhane Tsehay (Yemaneberhane@ethionet.et)

Version: 4 Date: 4 October 2011

Author's response to reviews: see over
Dear Associate Editor,

Thank you for the comments for the manuscript MS: 1675242103493150 “Oral and Anal Sex Practices among high School Youth in Addis Ababa, Ethiopia.” Thank you so much for the constructive, valuable and educative comments. I thank you very much, I learned a lot from the review process.

Please see below for a point-by-point response to your comments.

With best regards,

Amsale Cherie
Corresponding Author

Point-by-Point Response:

Comment 1: There is one major issue that needs to be resolved: The authors seem to have completely removed the limitation paragraph. I think my comments regarding this might not have been clear. What I wanted was (1) the limitation paragraph should not be the last one and (2) it needed to be revised. Specifically, the authors needed to discuss (1) the implications of the limitations (e.g., over- or underestimate associations) and (2) how your conduct, design or analysis of the data may have ameliorated the impact of these biases.

Response: Thanks, as per the suggestion we tried to discuss the strategies used to ameliorate the bias. Please see page 15.
Comment 2: The sample size should be reported in each table.

Response 2: sample size is indicated in all the tables.

Comment 3: The manuscript need to be copy edited for English grammar.

Response 3: The manuscript is copy edited by English speakers and comments incorporated.